Author’s response to reviews

Title: Evaluation of FTIR Spectroscopy as a diagnostic tool for lung cancer using sputum

Authors:

Paul D Lewis (p.d.lewis@swansea.ac.uk)
Keir E Lewis (keir_lewis@hotmail.com)
Robin Ghosal (ghosalr@hotmail.com)
Sion Bayliss (474488@swansea.ac.uk)
Amanda J Lloyd (abl@aber.ac.uk)
John Wills (438626@swansea.ac.uk)
Ruth Godfrey (a.r.godfrey@swansea.ac.uk)
Philip Kloer (Philip.Kloer@carmarthen.wales.nhs.uk)
Luis Mur (lum@aber.ac.uk)

Version: 5 Date: 23 November 2010

Author’s response to reviews:

Table has been inserted into manuscript

Authors details checked.

Tracking removed